Author's response to reviews

Title: Diagnostic and Pathogenetic role of Cafe-au-lait macules in Nevoid Basal Cell Carcinoma Syndrome

Authors:

Giovanni Ponti (giovanni.ponti@unimore.it)
Aldo Tomasi (altom@unim.it)
Lorenza Pastorino (lorenza.pastorino@unige.it)
Cristel Ruini (cristelruinni@gmail.com)
Carmelo Guarneri (carmeloguarneri@gmail.com)
Victor Desmond Mandel (victor.desmond.mandel@gmail.com)
Stefania Seidenari (stefania.seidenari@unimore.it)
Giovanni Pellacani (giovanni.pellacani@unimore.it)

Version: 4 Date: 25 October 2012

Author's response to reviews: see over
Dear Editor,

We would like to resubmit our manuscript entitled “Diagnostic and pathogenetic role of Café-au-lait macules in Nevoid Basal Cell Carcinoma Syndrome” by Ponti G. et al., corrected and modified according to the suggestions and criticisms of the Reviewer for publication in “Hereditary Cancer in Clinical Practice”.

We thank very much the Assessors and the Editorial Board for their helpful suggestions, which contributed to improve the scientific quality of our paper.

We therefore would like to resubmit the revised version of our manuscript.

Looking forward to hearing from you, best personal regards.

Sincerely yours

Dr. Giovanni Ponti
Assistant Professor of Clinical Pathology
Department of Clinical and Diagnostic Medicine and Public Health
Via del Pozzo, 71
41100 Modena, Italy
giovanni.ponti@unimore.i